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Not Activated
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Not Activated
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Not Activated

TBB2

_

Not Activated

TBC1

MBBL3090 (run 2)

Cold test to start

Cooling

TBC2

SSS06 removed; SSS02 will come

300 K

TBD1

MBBL3087

Disconnection

300 K

TBD2

MBBR2050

Cold test

1.9 K

TBE1

MBBR3019

Warming

Warming

TBE2

MBBR3077

Leak tests.

300 K

TBF1

MBAL2060

Cold test, Training

1.9 K

TBF2

MBAL2046

Leak tests.

300 K

SM18 Operation SCHEDULE Meeting (Tuesdays)

 Bench status  and tests for this week : 


Tested : 3031 (TBE2);3019 (TBE1);3087 (TBD1);3064 (TBF2); 3090 (run1)

· Priority list of magnets to be tested available in the cryodipole web page. To come : 2027 retested and 1047 (w.o anticryostat).

· Shafts in 3090 (C1); 206 (TBF1); 2050 (TBD2) then in 2046 (TBF2).

Test Coordinator this week:  Marco 160621 ;  MTM  Tech. on call Ewald:  (16-3602) . 

· Situation From Cryogenics:  J-P Lamboy or ACR Rep.

1. Leaks: 3090 (TBC1) bellows; 3019 (TBE1) Joint or detection system,  3031 (TBE2) Joint or detection system.
2. Wednesday night: Lost of communication in PLC of the CFB. CFB not controlled. Restart Thursday morning.
3. Friday : stop of the CWS for reconditioning ( loss to high)
4. TBF2 (3064) : Discrepancy between TT821 (302 K ) and Tcryo (170 K).
· Other General Issues
1. TBD1 (1037) : Insulation  dipole vs. ground, good at warm, failed below 250 K. End of the story.

2. SSS06: Short between sextupole and octupole correctors

3. General water cut : Wednesday afternoon.

4. 3090: Thermal cycle executed without going to 9 T or executing  9 ramps.

	Date
	Problem
	Type
	Application/soft.
	Action/advise

	Wednesday_18_A


	TBD1 (3087): IAP@warm : energy  of 1 LF QH less than the limit
	MTM
	IAP
	Giorgio informed.

Fit in IAP to be improved.

Done by Boris.

	Wednesday_18_N


	TBD2: No magnet names in TEMA
	MTM
	TEMA
	Solved by Giorgo.  Scan and run of test .

	Wednesday_18_N


	TBD2: No Temperature data for TT821

IAP not working properly in Sun MTA37
	AB/CO
	IAP
	Advise of Giorgio:

LF DAQ slot 6 checked : T=279 K.

Boris informed.

Use : another machine for analysis



	Thursday 19_N


	TBE2: change of Iearth during the quenches.
	MTM
	
	Magnet maintained at 1.9 K.

Pierre’s instruction: HV test performed : OK

Additional quench performed. OK then warm up

	Friday 20_A


	TBC1: DAQ not getting armed for IAP

Abort and re-launched. No success.
	AB/CO
	
	Solution : Reset  VME crates then DAQ got armed.

	Friday 20_N

Sunday 22_A


	TBF2: Slow PA launched. Failed. QH discharged.

TBD2: same problem.
	MTM
	
	Giorgio advised to check whether the 2 A cable was connected to the magnet.  Check all the connection: Found that D20 not connected.

	Saturday 21_A


	TBF2 (3064): Discrepancy between reading of TT821 (302 K) and Tcryo (170 K)
	MTM/ACR
	
	

	Saturday 21_A


	TBD1: Problem with AC compensation of potaim cards D1-L-U,D1-D2
	MTM
	IAP
	

	Saturday 21_A


	TBE2 (3077): no anticryostat stickers “no Anti Cryostat”is not recognised by the system during the scan.
	MTM
	HWREC
	Solution proposed by Giorgio: leave the field empty.

	Saturday 21_N


	Problem with warning boxes”high currents” in particular for D1.
	MTM
	
	Action launched by Giorgio: Improvement of the fixation of these boxes.

	Sunday 22_A


	TBE2 (3077): Name in the sticker on the magnet wrong : impossible to start TEMA.
	MTM
	
	Giorgio came and corrected the name.

	Sunday 22_N


	TBF1: Sun MTA 28 blocked.
	MTM
	
	

	Sunday 22_N


	TBE2: PT10 Min energy :LF DAQ armed but not the correct leds are glowing

Probable origin : no exit for all applications in particular LF web.
	MTM
	
	Advise by Giorgio : reset and re-launch.

	Monday 23_M


	TBE2: Unable to put 2 A for warm up.

Jack to be replaced in cable connected to the current leads.
	MTM
	
	Advises :

-Be gentle with the cables: remove it using the connectors.

- If something broken : pass the message in order to replace it asap.








